Author’s response to reviews

Title: COMMUNICATION SKILLS INTERVENTION: PROMOTING EFFECTIVE COMMUNICATION BETWEEN NURSES AND MECHANICALLY VENTILATED PATIENTS

Authors:

KS Dithole (oyedeoa@unisa.ac.za)

Gloria Thupayagale-Tshweneagae (tshweg@unisa.ac.za)

Oluwaseyi Akpor (akporoa@abuad.edu.ng)

Mary Moleki (Molekmm@unisa.ac.za)

Version: 4 Date: 23 Nov 2017

Author’s response to reviews:

AUTHOR’S RESPONSE TO REVIEWS

Title: COMMUNICATION SKILLS INTERVENTION: PROMOTING EFFECTIVE COMMUNICATION BETWEEN NURSES AND MECHANICALLY VENTILATED PATIENTS

Authors:

K.S. Dithole (Ditholek@mopipi.ub.bw)

Gloria. Thupayagale-Tshweneagae (tshweg@unisa.ac.za)

Oluwaseyi A. Akpor (akporoa@abuad.ac.za)

Mary M. Moleki (molekmm@unisa.ac.za)

Version4: Date: 23/11/2017

Authors’ response to Editorial review:

1. In your “ethical approval and consent to participate” section please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

The information has been included
2. In your “ethical approval and consent to participate” section please confirm whether written consent was obtained from participants. If written consent was not obtained, please explain why not.

The information has been included

3. In your “consent for publication” section of your declarations, please confirm whether your participants gave consent for direct quotes from their interviews to be used in this manuscript. Please include a statement to this effect in this section of your declarations.

Done

4. Please remove the response letter from your supplementary files.

Done

5. We note that the current submission contains some textual overlap with other previously published works, in particular:

Dithole, K. S. (2014). Nurses' communication with mechanically ventilated patients in the intensive care units (Doctoral dissertation). This overlap is spotted throughout your manuscript.

Done